Weekly Status Week of September 11 – September 17, 2008

Day of Year 255 - 261

Mission Day 2859 - 2865

Earth Observing-One (EO-1) General

There were 117 Data Collection Events (DCEs) scheduled this week.  

INSTRUMENTS

All instruments operated nominally this week.

Solar calibration:

· Hyperion solar calibration was performed on September 17, 2008 at 11:43z.

Lunar calibration:

· An all-instrument nominal lunar calibration was performed on September 15, 2008 during the 21:10z umbra.

· A modified lunar calibration was performed two orbits after the nominal calibration.

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders. Engineering data is being received during real-time contacts and for all science imaging events.

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

A face to face meeting was held between members of the GSFC SensorWeb team, GSFC Management, NRO, and DIA on Sept 11 that was attended by the Director of NRO Futures Lab, the DIA Materials and Signatures Intelligence Program Director and the Division Chief of the DIA Advance Technology Division of the Materials and Signatures Intelligence Directorate.  The cross-domain software status and plans were discussed.  In particular, we discussed possible scenarios involving humanitarian assistance for emergencies such as floods, wildfires, etc.  The next full meeting of the focus group is scheduled for early-mid October.

An effort will be made to have Everett Hinkley (Forest Service) give a briefing to NRO/DIA/NGA on how the Forest Service manages the data acquisition and distribution for response to wild fire emergencies.

Sensor Web & Virtual Observatory Demonstrations

A teleconference between the Advanced Information System Technology (AIST) Sensor Web collaborators was held on Thursday, September 11.  The following items of interest were discussed:

· The previously scheduled two Ikhana UAS wildfire related flights were reduced to one flight to occur on September 19 and to focus on a wildfire in the Mt Shasta region of Northern California.

· The team will revise the triggering software to deliver a target to the Ikhana SPS from the new GSFC Campaign Manager.

· The EO-1 took one image of a wildfire in Northern California on Sept 17 and delivered the data for use as a potential trigger for the Ikhana flight on Sept 19.

· This potential trigger was also delivered to Draper on Sept 18 to develop a new Ikhana flight plan.

· The team agreed to work towards having Ames deliver a trigger to the EO-1 Campaign Manager based on an Ikhana observation.  A new API for the Campaign Manager was developed and delivered to the team to enable automation of this API.

· The GSFC Mission Science Office (MSO) has tentatively agreed to turn on the automatic scheduling service based on the new Campaign Manager for two months as a trial starting Oct 1.  The MSO has been trained on use of the Campaign Manager to schedule their baseline scenes.

· The Northup Grumman smoke forecast team successfully posted a new trigger (lat/lon) to the new Campaign Manager API based on smoke forecast from the BlueSky model using OpenID security.

· A new feature for the GSFC Campaign Manager was developed that displays EO-1 flight tracks and target swaths on Google Earth in KML format.

· WVHTC Foundation is revising their SensorML workflow for Hyperion data to include fire products from the JPL web processing service (WPS).

· The GSFC team is implementing the ALI Level 1G processing and delivery of the sensor observation service (SOS) Geo-located band files.

Another teleconference with the flood early warning sensor web collaborators was held on Tuesday, September16 to firm up plans for the CEOS WGISS meeting in Boulder, CO on 24 September, collaboration activities for the GEOSS AIP meeting also in Boulder on 25-26 September, and the UN-SPIDER meeting in Bonn, Germany on 12-13 October.

The GEOSS AIP team also had a teleconference later on September 16 to further finalize its agenda.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week with the exception of the period when the lunar calibration was performed.

EO-1 MISSION OPERATIONS CENTER

Real-Time

Spacecraft state of health is currently nominal.

Command Management System (CMS) upgrade to Stored Command Processor (SCP):

· Preliminary testing, comparing command loads generated by the new ASIST/SCP and the old CMS has begun 

· FOT is working on a scheme to test all commands in the command database

GPS leap second:

· FOT has noticed a slight step increase in the difference of cross-track component of the orbit

· The timing of the step increase coincided with the upcoming leap second being broadcasted over the GPS satellite network

· FOT introduced an offset in the EOP (Earth Orientation Parameters) uplinked to the spacecraft every Sunday from the MOC

· Offset resolved difference in cross-track data 

· This offset will be removed from the system after January 1, 2008 when the leap second is officially introduced to all systems.

Mission Planning

ASPEN upgrade:

· Currently installing MATLAB on ASPEN machines.

· Systems administrator currently working closely with Mathworks (MATLAB developers) to resolve issues

· MATLAB functioning in command-line mode, but not through GUI.

· FOT has received IDL software

Flight Dynamics

FOT continues to perform parallel operations between autoProducts and XFDS:

· To date, only two issues observed

· Some files created on one machine and not the other

· Difference in grazing angle causing occasional disagreements regarding GN/SN visibility (typically on order of 1 – 4 seconds difference)—most likely caused by STK update to full WGS84 earth model 

PF2 tracking data problems:

· Tracking data provided from the station can not be used due to bad Doppler measurements.

· FOT has provided in-depth analysis of tracking data difficulties

· Observing large spikes in observed relative velocity

· Also observing systemic error in observed relative velocity which is most likely not simply due to a simple bias

Tracking data accounting:

· FOT is now being notified if a station fails to deliver tracking data post contact

· FOT currently notifying ground stations when tracking data is not received

System Administration

MATLAB installation on ASPEN machines:

· MATLAB currently working only in text mode on the machines.

GROUND AND SPACE NETWORK

Station Downtimes

McMurdo Ground Station, AK, US

· Changed status to GREEN.

· McMurdo supports will be included in the schedule beginning September 29, 2008.

· A Network Advisory Message (NAM) has announced that McMurdo’s 10m antenna will be down for an antenna control unit upgrade from October 20, 2008 to January 15, 2009.

Operational Discrepancies

No major problems to report.

Upcoming Events

Mean Local Time (MLT) control inclination change maneuvers:

· First 400-s inclination maneuver on Sep 30, 2008 at 13:01:09z

· Second 400-s inclination maneuver on Oct 02, 2008 at 14:18:12z

· Projected to raise MLT above 10:00 am.

Imagery Status

Scenes and Engineering Cals planned for week of September 11 – September 17, 2008              117

Total scenes and engineering calibrations planned for entire mission – approximately 42,167

Total Scenes:  ALI scenes in the Level 0 archive            38,023 (as of September 17, 2008)

                        Hyperion scenes in the Level 0 archive    37,773
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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